

Re: Answers for Reviewers Comments

Sternoclavicular Joint Dislocation and its management – A review of literature

Manuscript NO.: 20075

Dear Editor

Thank you for your recent e-mail with the reviewer's comment.

The following changes have been made to the manuscript

1. The title has been amended to "Sternoclavicular Joint **Dislocation** and its management – A review of literature
2. Page 8
"CT imaging is readily available 24/7 in most trauma units and this is the investigation of choice"
Changed to
'CT imaging is readily available **24 hours a day** in most trauma units and this is the investigation of choice
3. Key words – Physiotherapy has been replaced with '**Surgery**'
4. Page 4 – CLL abbreviated '**costoclavicular (rhomboid) ligament (CLL)**'
5. Page 5 **Spencer et al^[6]**
6. Page 7 – Location of **figures** included in the text
7. **References** have been rearranged as per guidelines
8. Figures have been submitted in powerpoint slides
9. **Tracking** changes accepted
10. **Audio core tip** provided.

Please let us know if you need further information.

Kind Regards

David Thyagarajan

Daniel J Morell